**Author’s response to reviews**

**Title:** Requiring smartphone ownership for mHealth interventions: Who could be left out?

**Authors:**

Krishna Bommakanti (kbommakanti@ucsd.edu)

Laramie Smith (laramie@ucsd.edu)

Lin Liu (l2liu@ucsd.edu)

Diana Do (dianado26@yahoo.com)

Jazmine Cuevas-Mota (jcuevasmota@ucsd.edu)

Kelly Collins (Kelly.Collins@sureadhere.com)

Fatima Munoz (fmunoz987@gmail.com)

Timothy Rodwell (trodwell@ucsd.edu)

Richard Garfein (rgarfein@ucsd.edu)

**Version:** 2  **Date:** 28 Oct 2019

**Author’s response to reviews:**

1. Consent to participate

   Please copy the statement regarding written consent to participate from the Methods section to the "Ethics approval and consent to participate" subsection of your Declarations

   We have copied the statement below and included it in the ‘Ethics approval and consent to participate’ section of the Declarations.

   “Written informed consent was obtained from all participants and the study protocol was approved by the Institutional Review Boards of the University of California San Diego, University of California San Francisco, the New York City Department of Health and Mental Hygiene, and the San Diego County Health and Human Services Agency Research Committee.”

2. Role of Funders

   Please note the role(s) of the funding bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

   We have included the following statement to clarify the role of funding agencies in this study.
“The funding agencies had no role in the design of the study; data collection, analysis, or interpretation; or writing the manuscript.”

3. Author initials
In the “Authors’ contributions” section of the Declarations, please ensure initials match their provided names in the author list for this manuscript, in the correct order. For instance, an author named John Steven Doe, their initials should be represented as JSD in the “Authors’ contributions”; an author named Jane Ann Doe should be referred to as JAD.

We have ensured that all initials in this section match the names provided in the author list.

4. Authors’ contributions
Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript in its current state. Currently, not all authors are listed as having approved the final draft of the manuscript. It is essential this is clearly stated for all contributing authors.

We have modified this section to indicate that each author approved of the final draft of the manuscript with the statement below.

“KKB conceived of the study, analyzed data, interpreted study findings, drafted the manuscript, and prepared the final version of the paper for publication. RSG contributed to the study design, analysis of findings, manuscript drafting, and final approval of manuscript for submission. LRS and TRC contributed to interpreting findings, manuscript drafting and final approval of manuscript for submission; LL oversaw data analysis, contributed to manuscript drafting and review, and approved the final manuscript for submission; DD contributed to data analysis, manuscript drafting, and final approval of manuscript for submission; JCM, oversaw data collection and management, contributed to manuscript drafting, and approved final approval of manuscript for submission; KC and FM contributed to manuscript drafting and final approval of manuscript for submission.”

5. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

All track changes have been removed and the text color is uniform throughout the manuscript.